Author’s response to reviews

Title: Online Cost-Effectiveness ANalysis (OCEAN): A user-friendly interface to conduct cost-effectiveness analyses for cervical cancer

Authors:

David Moríña (dmorina@mat.uab.cat)

Jose Ignacio Martí-Gil (jimartigil@gmail.com)

Pedro Puig (ppuig@mat.uab.cat)

Mireia Diaz (mireia@iconcologia.net)

Version: 2 Date: 05 Jul 2019

Author’s response to reviews:

Dear editor,

We have reviewed the manuscript and avoid overlapping text with our previous publications wherever it was possible (according to Urkund anti-plagiarism software the overlapping is about 12%). However, we think that it should be taken into account that the example used in this manuscript to illustrate the usage of the OCEAN tool was previously analysed in one of our publications (for comparability purposes) and therefore some overlapping is unavoidable.